
\documentclass[11pt]{article} 

\setlength{\oddsidemargin}{0in}
\setlength{\evensidemargin}{0in}
\setlength{\topmargin}{0.0in}
\setlength{\headheight}{0in}
\setlength{\headsep}{0in}
\setlength{\textwidth}{6.5in}
\setlength{\textheight}{9.0in}
\setlength{\parindent}{0in}
\setlength{\parskip}{0.1in}

\usepackage{times}
\usepackage{fancyvrb}  
\usepackage{relsize}
\usepackage{hyperref} 

\begin{document}

\title{Introduction to Thrust Programming: a Tutorial}

\author{Norman Matloff\\
University of California, Davis}

\date{}  

\maketitle

My tutorial on Thrust programming is a (more or less independent) 
chapter in my open-source textbook on parallel programming, at
\url{http://heather.cs.ucdavis.edu/~matloff/158/PLN/ParProcBook.pdf}

\end{document}


